Author’s response to reviews

Title: Establishing partnership with Traditional Birth Attendants for improved maternal and newborn health: a review of factors influencing implementation

Authors:

Tina Miller (tamiller@brookes.ac.uk)
Helen Smith (helen.smith@lstmed.ac.uk)

Version: 3 Date: 23 Sep 2016

Author’s response to reviews:

A 'Declarations' section has been added to the paper as requested and as below:

Ethics approval and consent to participate
Not applicable

Consent for publication
Not applicable

Availability of data and material
Not applicable

Competing interests
The authors declare that they have no competing interests

Funding
The authors did not receive any funding for the production of this article
Authors' contributions

TM produced the first complete draft of the manuscript. HS revised the manuscript and both authors edited, read and approved the final version

Acknowledgements

The authors wish to thank Annie Portela of the Department of Maternal, Newborn, Child and Adolescent Health, WHO, for her helpful feedback on earlier drafts of the paper and for editing the series of which this paper is part